

ISSN 2385-2755 DiSSE Working papers [online]

WORKING PAPERS SERIES DIPARTIMENTO DI SCIENZE SOCIALI ED ECONOMICHE

A Beam of Light: Media, Tourism and Economic Development

Samuel Nocito, Marcello Sartarelli, Francesco Sobbrio



SAPIENZA - UNIVERSITY OF ROME P.le Aldo Moro n.5 – 00185 Roma T(+39) 0649910563 CF80209930587 – P.IVA 02133771002

A Beam of Light: Media, Tourism and Economic Development^{*}

Samuel Nocito[†]

Marcello Sartarelli[‡]

Francesco Sobbrio[§]

August, 2021

Abstract

Tourism accounts for around one tenth of global GDP. We analyze the impact of entertainment media in drawing tourists to filming municipalities (*media multiplier*) and, in turn, the effect of tourism on local economic development (*tourism multiplier*). To quantify the *media multiplier*, we employ a triple-difference empirical strategy exploiting the staggered international release across the EU of *Inspector Montalbano*, a TV series set in four municipalities of Sicily, a region of Italy. We find that the series release led to a fourfold increase in the number of tourists and boosted tourist expenditure by a factor of 2.5. Furthermore, we provide evidence of positive spillovers in nearby municipalities. To estimate the *tourism multiplier*, we exploit the interaction between the filming locations and the time-varying share of countries in which the series was aired, to instrument total tourist expenditure at the municipality-time level. Our results show that a 10% increase in total tourist expenditure translates into an increase in municipal income of 4.7%. The paper suggests that both entertainment media and tourism can be effective tools to boost local economic development.

Keywords: Economic Development, Media, Spillovers, Tourism **JEL Classification**: L82, L83, O12, R10, Z32.

^{*}We are grateful to *Palomar* S.p.A. for providing us with the information on the international release dates of the *Inspector Montalbano* TV series. Rebecca Pisicchio and Giovanni Rillo provided excellent research assistance. We would also like to thank Marit Hinnosaar, Eliana La Ferrara, Hannes Muller, Guido Savignano, Mara Sorella, Yi Zhu and seminar participants at CUNEF, Loyola University, Universidad Complutense de Madrid, and Luiss University for helpful comments and discussions. The usual disclaimers apply.

[†]Luiss University, snocito@luiss.it

[‡]Universidad Complutense de Madrid and ICAE, msartare@ucm.es

[§]University of Rome Tor Vergata & CESifo, Francesco.Sobbrio@uniroma2.it

1 Introduction

Tourism accounts for around 10% of global GDP and employment (World Travel Tourism Council, 2019). However, it is unclear whether and to what extent it can be an effective tool for boosting economic development. A thorough quantitative analysis on the economic impact of tourism is empirically difficult: tourist flows and expenditure are likely to be affected by endogenous choices at the local level such as tourism investment. Similarly, while there is considerable anecdotal and correlational evidence suggesting that entertainment media can be a powerful means of attracting tourism in filming locations (e.g., Riley *et al.* 1998; Hudson and Ritchie 2006b), drawing causal inference on the link between entertainment media and tourism remains challenging.¹ An adequate counterfactual is typically lacking as media locations and the timing of releases are potentially endogenous.

This paper addresses these challenges by exploiting the staggered international release of In-spector Montalbano, an Italian TV series, whose main character is a police detective living in a fictional seaside town in Sicily. The municipalities in which the series was filmed (which we term "Montalbano municipalities"), were located in the province of Ragusa in an area historically little known to tourists. This offers an ideal setting to assess the causal impact of entertainment media on tourism. Specifically, we use a triple-difference empirical strategy that exploits the differences in the average value of tourism outcomes (number of tourists or tourist expenditure): i) between countries exposed to the TV series (treated) and those not exposed (control); ii) between Montalbano municipalities and other municipalities in the same Italian region (Sicily); iii) before and after the release of the series. In turn, the exogenous shock created by the series on tourist expenditure provides us with a unique opportunity to estimate the causal effect of tourism on economic development. Specifically, we instrument total tourist expenditure at the municipality-time level using the interaction between the time-varying share of countries exposed to Inspector Montalbano and a dummy identifying Montalbano municipalities.

The analysis leverages a linked dataset, created by merging three data sources. The first is a

¹A noteworthy–if anecdotal–example may be found in the success of Petra, Jordan, as a tourism destination, which is usually attributed to the 1989 movie *Indiana Jones and the Last Crusade* (Business Insider, 2019). Other famous examples include the reputed links between *The Lord of the Rings* and tourism in New Zealand (Forbes, 2012) or *Braveheart* and tourism in Scotland (BBC News, 2015). Several contemporary TV series are believed to have had a similar impact, such as *Breaking Bad* in New Mexico (USA Today, 2013) and *Games of Thrones* in Ireland, Spain, Iceland and Croatia (Telegraph, 2019).

rich survey dataset from the Bank of Italy on international tourist flows containing information on the date of the visit, country of origin, municipalities visited, and money spent in each visited municipality for the period of 1997-2017 for tourists from all European Union countries.² The second source is comprised of administrative data from the Italian Ministry of Economy and Finance on personal income at the municipality-year level. Finally, we also collected data on the number of workers and firms in the tourism sector from the Italian National Statistical Institute, similarly available at the municipality-year level.

In terms of *media multiplier* (Besley *et al.*, 2020), our baseline estimates show that the number of tourists visiting *Montalbano* municipalities increased fourfold after the release of the TV series, whereas tourist expenditure increased by a factor of 2.5. Importantly, we document that the series release did not crowd out tourism from other municipalities in the same region. Indeed, it appears to have created positive spillovers, especially in nearby municipalities.

For what concerns the impact of tourism on local economic development, we first document that an increase in total tourism expenditure leads to an expansion in the supply of touristic services at the municipal level (i.e., number of firms and workers operating in the tourist sector). A 10% increase in total tourist expenditure translates into an 11.5% increase in the number of firms and a 10.1% increase in the number of workers in the tourism sector. We then quantify the *tourism multiplier*: our estimates indicate that a 10% increase in total tourist expenditure translates into a 4.7% increase in municipal income. We also show that the latter effect is driven by an increase in the extensive margin (more taxpayers overall) as well as the intensive margin (more income per taxpayer). Furthermore, the *tourism multiplier* effect is also associated with a shift of taxpayers from lower to higher income brackets.

We present several alternative specifications that demonstrate the robustness of our baseline results. When looking at the impact of media on tourism, our alternative specifications include (among others) the exclusion of larger municipalities, limiting our sample to the countries in the European Union at the time of the first international release of *Inspector Montalbano* and using an alternative dataset drawn from *Tripadvisor*.³ For our analysis of the impact of tourism on

²Specifically, we consider all international tourists coming from any of the 27 countries belonging to the European Union in 2017, which is when our sample period ends. Around 1/3 of the countries in our sample had been exposed to the *Inspector Montalbano* TV series by 2017.

 $^{^{3}}$ We limit the use of the *Tripadvisor* data to a robustness check since it covers a more limited time period and does not contain information on tourist expenditure.

economic outcomes, we provide robustness checks controlling for linear or quadratic time-trends at the local level, focusing on different sub-samples or exploiting the use of alternative instruments (e.g., weighting the share of countries exposed to *Inspector Montalbano* by country population or GDP).

Most importantly, the paper presents an extensive set of tests to formally validate our empirical strategy. The identifying assumption behind our triple-difference empirical strategy is that relative differences in tourism outcomes between treated and control countries trend similarly in Montalbano and control municipalities prior to the series release (Olden and Møen, 2020). Accordingly, we provide evidence to support this assumption of parallel trends in tourist inflows and expenditure. While this would suffice to corroborate the validity of our empirical strategy, we present additional evidence that demonstrates that the study setting exhibits several other desirable features. In particular, balance tests show that the observable characteristics of *Montal*bano and control municipalities are comparable in pre-treatment years. Similarly, we show that these observable characteristics cannot predict which municipalities would go on to be treated (randomization test). Further, using data on Google searches (in tourists' countries of origin), we demonstrate that there are no pre-trends of interest in the filming locations or the TV series prior to its international release. At the same time, we complement our baseline specification with an event-study approach that documents the absence of pre-trends in tourist inflows and expenditure. In addition, we present random inference tests in the spirit of Young (2018) and Dell and Olken (2020) that provide counterfactual placebo estimates of our treatment by randomly assigning a treatment to each country-municipality-year-month combination and comparing them to our baseline estimates. Finally, to address possible concerns related to the staggered nature of our treatment (Goodman-Bacon, 2018; Callaway and Sant'Anna, 2020; De Chaisemartin and d'Haultfoeuille, 2020; Athey and Imbens, 2021), we present estimates that consider a single cohort of treated countries at a time, with never-treated countries as controls.

Our paper contributes to the literature on the economic impact of media (see DellaVigna and La Ferrara 2015 for a review) by quantifying the economic impact on tourism of a credibly exogenous variation, across countries and time, in the exposure to entertainment media showing attractive but otherwise unknown tourist locations. In this respect, the closest papers to ours are Hinnosaar *et al.* (forthcoming) and Besley *et al.* (2020) on the effects on tourist inflows of Wikipedia and

news on violent events, respectively.⁴ In this regard, our paper offers insights on the impact of entertainment media rather than that of additional information (Hinnosaar et al., forthcoming) or negative news content (Besley et al., 2020). This allows us to consider different and more direct policy implications relating to media and tourism. Moreover, by exploiting within-region variation in the treatment, we can narrowly identify the causal effect with our triple-difference strategy as well as assess the presence of potential spillover effects across tourist locations.⁵ Our findings on the impact of entertainment media on tourist destinations also speak to the literature on advertising and consumer persuasion (see DellaVigna and Gentzkow 2010 for a review). Stephens-Davidowitz et al. (2017) show that a 100 percent increase in Super Bowl ad impressions (defined as an increase of 100 Nielsen rating points) increases the revenue of featured movies by 50-70%. Similar elasticities have been found by Hartmann and Klapper (2018) in a study of beer and soft drink advertisements. Exploiting a randomized mail field experiment in South Africa, Bertrand et al. (2010) demonstrate the importance of advertising content, noting that certain features may lead to large effects. Valletti and Veiga (2021) find that looking at an online ad for an extra second increases the probability of purchasing a product by 0.72%. Our setting differs from these studies in terms of the advertising object: travel destinations rather than traditional consumer goods or services. More importantly, our setting also differs in terms of the advertising message, which is implicit and embedded within a media entertainment product.⁶

Our analysis complements studies of the impact of tourism on economic development such as Faber and Gaubert (2019), who test predictions from a general equilibrium model on the effect of tourism on development at the local and national level by exploiting pre-existing differences

⁴Our findings are also related to the literature on marketing and tourism, which has focused on the role played by specific movies in influencing tourist flows (Busby and Klug, 2001; Hudson and Ritchie, 2006a), the characteristics of movies that are positively associated with tourist flows (Riley and Van Doren, 1992; Riley *et al.*, 1998; Hudson and Ritchie, 2006b; Beeton, 2016), or stated preferences for a tourist destination (Soliman, 2011). Our paper contributes to this literature by highlighting the importance of exploiting credibly exogenous variation in exposure to entertainment media to properly infer causal effects.

⁵In contrast to Hinnosaar *et al.* (forthcoming), we document an impact of media on tourism inflows beyond the intensive margin. In their examination of the Wikipedia treatment, Hinnosaar *et al.* (forthcoming) do not see any significant impact on Google searches, implying that their effect works through the intensive margin (e.g., among people already interested in a given locality, looking at Wikipedia to decide whether it is worth visiting or not). Instead, we observe a positive and significant impact of the exposure to entertainment media on Google searches, suggesting that media exposure also affects the extensive margin.

⁶The empirical link we uncover regarding media, tourism and local development also provides micro-level evidence in support of the macroeconomic literature suggesting a positive impact of advertising on economic activity at the aggregate level (Taylor and Weiserbs, 1972; Chowdhury, 1994; Seldom and Jung, 1995; Grossmann, 2008; Molinari and Turino, 2018).

in attractiveness between Mexican municipalities (e.g., beaches and historical sites).⁷ Our study differs both in terms of its economic context (an underdeveloped region of a developed country in our case, as opposed to a developing country in Faber and Gaubert 2019) and its identification strategy, as we use a partial equilibrium empirical model and exploit a time-varying instrument for tourist expenditure based on the staggered international broadcast of a television series. We can therefore control for all time-invariant characteristics at the municipality level and are able to provide evidence on the absence of pre-trends.

Finally, this paper also relates to studies on within-country regional development (e.g., Tsui 2007 on China; Lintz and Schmude 2005; Heise and Porzio 2019 on Germany, Martin 1988; Gardiner *et al.* 2013 on Great Britain; Ballarino *et al.* 2013; Federico *et al.* 2019 on Italy). These studies suggest that the determinants of regional divides range from historical or institutional differences to resilience to a number of region-specific shocks.⁸ Our paper contributes to the regional development literature by looking at Sicily, an underdeveloped region of a developed country (Italy) in which industrial policies and public subsidies have not been particularly successful (Accetturo and De Blasio, 2019; Albanese *et al.*, 2020). Our findings suggest that entertainment media and, in turn, tourism might be powerful engines of development that have been overlooked in the literature.⁹

The remainder of the paper is structured as follows. Section 2 describes the characteristics of Sicily and of the TV series that are relevant for our empirical analysis. Section 3 presents the data and Section 4 outlines the empirical strategy. Section 5 discusses the main results on the effect of entertainment media on tourism, along with robustness specifications and validity tests. Section 6 introduces evidence on spatial spillovers, while Section 7 provides the results on the effect of tourism on economic development. Finally, Section 8 concludes.

⁷For cross-country evidence on the determinants and effects of tourism see Eilat and Einav (2004); Sequeira and Maçãs Nunes (2008); Arezki *et al.* (2009); McGregor and Wills (2017).

⁸Becker *et al.* (2013, 2018) analyze the impact of regional funds allocated by the European Union to mitigate economic disparities and show that while this funding has fostered convergence, the speed at which the latter takes place seems to depend on predetermined economic conditions in a given region. See also Albanese *et al.* (2020) for the case of southern Italy.

 $^{^{9}}$ For other studies looking at various factors influencing local development in the context of Italian municipalities see Bordignon *et al.* (2003); Barone and Mocetti (2014); Galletta (2017).

2 Background: Sicily and Inspector Montalbano

In this section we describe the main characteristics of Sicily and the *Inspector Montalbano* TV series that are relevant to understanding our empirical strategy and interpreting the results.

2.1 Sicily

Sicily is an island region in the far south of Italy. It is home to around 5 million inhabitants, more than Ireland or Croatia and similar to US states such as South Carolina or Colorado. Measuring around 26,000 square kilometers in size, Sicily is larger than Slovenia or US states such as Vermont or New Jersey, and accounts for 5.2% of Italy's GDP. In PPP terms (2016, US\$), GDP per capita in Sicily is around \$24,300, the second-lowest in the country and slightly above that of Cyprus. For comparison, average Italian GDP per capita is 1.6 times higher, while Mexico's is around 1.3 times lower (World Bank, 2020).

The persistence of the Italian north-south divide in economic development has attracted considerable attention from both academics and policymakers. This issue is not, however, specific to Italy but also applies to other developed countries, to varying degrees. One such example is Germany (D'Amuri *et al.*, 2010; Heise and Porzio, 2019), which is characterized by regions that lie far behind the national average, such as the Ruhr. Different solutions have been offered to solve this issue. A first one is a massive relocation of workers to more prosperous regions: indeed, Italy experienced extensive internal migration from the south to the north after WWII (Bartolucci *et al.*, 2018). Nevertheless, there are large costs associated with the relocating of people across space (Becker *et al.*, 2004; Anelli and Peri, 2017; Heise and Porzio, 2019). A clear alternative is to foster the economic development of disadvantaged regions. In Italy, these efforts have taken the form of subsidizing firms willing to open factories in the south and by injecting vast amounts of public funds, though with little success (Accetturo and De Blasio, 2019; Albanese *et al.*, 2020).

Others, meanwhile, have noted the South's considerable tourism potential and attractiveness, advocating for these regions to foster tourism as a way to boost economic development. However, despite its mild climate, long coastline, and very rich cultural and archaeological heritage, tourism accounts for only 5.3% of Sicilian GDP in terms of direct contributions and 11.4% in terms of total contributions (Caselli, 2017).¹⁰ This is similar to the national average, but below several other

¹⁰It is worth adding that, despite having historically been the cradle of the Mafia (Bandiera, 2003; Buonanno

Italian regions in terms of tourism's share of total GDP (which is higher in Calabria, Sardinia, Puglia, Abruzzo, Emilia-Romagna, Veneto) as well as total GDP from tourism (higher in Campania, Emilia-Romagna, Lombardy). In Italy, average per capita added value generated by tourism is 3,000 euros-a figure that is only 1,800 euros in Sicily, one of the lowest in the country.

2.2 Inspector Montalbano

Inspector Montalbano is a TV series featuring a police detective (Montalbano), who lives in a fictional seaside town in Sicily. The series is set in various locations in four municipalities of the province of Ragusa (see Figure 1).¹¹ Figure 2 provides a map of European Union countries (excluding Italy) at the end of our sample period (2017), showing whether the country was exposed to *Inspector Montalbano* or not, as well as the year in which the series was released in each treated country.¹² The release year refers to the first broadcast of *Inspector Montalbano* on free-to-air TV channels. Such free-to-air broadcasts of *Inspector Montalbano* are likely to represent a supply-side shock in the broadcasting country. Specifically, the broadcasting of the TV series increases the supply of images of historical and natural landscapes and other amenities (e.g., food), among other characteristics, connected to the *Inspector Montalbano* locations.

In principle, one might be concerned by the potential endogeneity of such a supply shock. That is, treated countries and municipalities might be endogenous as the release of the series in a given country might be driven by pre-existing interest in visiting the destinations. While Section 4 formally presents our empirical strategy and provides evidence to support our identifying assumption, there are some reasons why such concerns are unlikely to be valid in this context.

First, Inspector Montalbano was not a typical "blockbuster" production. The per-minute production cost of each episode was around \$15,000 (which is around 1/8 of the cost of series such as Game of Thrones).¹³ Importantly, the key feature of the series is the crime-solving plot of each episode. As such, it is likely that international broadcasters chose to buy the broadcasting rights

et al., 2015), Sicily has a relatively low crime rate compared to many other Italian or EU regions (Eurostat, 2021). ¹¹The series is mainly set in five locations: Donnafugata, Modica, Puntasecca, Ragusa and Scicli, which are located in the Sicilian municipalities of Modica, Ragusa, Santa Croce Camerina and Scicli.

¹²The specific year-months of the official release of *Inspector Montalbano* across countries are as follows: France: 09/2000; Sweden: 06/2001; Germany: 09/2001; Finland: 11/2001; United Kingdom: 12/2008; Hungary: 01/2009; Spain: 01/2012; Greece: 09/2012. These release dates were kindly provided to us by *Palomar S.P.A.*, the production company for *Inspector Montalbano*, which holds the copyright to the TV series.

¹³ "Quanto Costano e come rendono le grandi fiction Rai e Sky", June 28, 2019; "How much money 'Game of Thrones' episodes cost to make in the final season, and throughout the series", *Business Insider*, April 19, 2019.

due to their assessment of possible demand for an Italian (or even Sicilian) crime drama rather than due to viewers' potential interest in a very specific area of Sicily.

Second, the municipalities where the series was staged were "off the beaten path". Thus, international viewers were unlikely to know them before they watched the series. To this point, Table 1 provides evidence showing that various demographic, geographic, economic and tourism characteristics assessed in census year 2001 (i.e., at the time of the series' first international release) are balanced between *Montalbano* municipalities and control ones. Moreover, in order to test whether *Montalbano* municipalities were chosen for a particular characteristic, we also perform a randomization test showing that none of the observable characteristics can predict which municipalities are treated. Both tests show that the *Montalbano* municipalities were rather similar to an average Sicilian municipality, and thus implicitly suggest that such municipalities were instead different with respect to the island's traditional tourist locations (e.g., Palermo, Taormina).

Google Trends data provide additional evidence on the orthogonality of the *Inspector Montal*bano release dates with respect to demand-side factors. Specifically, we gathered data from Google Trends on the volume of searches related to *Inspector Montalbano*, i.e., "Montalbano", or its location, i.e., "Ragusa", the main setting and the capital of the province where the series is filmed. The left panels of Figure 3 plot the average monthly index of Google Trends search volumes related to both of the above terms after regressing the volumes with respect to country and year-month fixed effects.¹⁴ The right panels of Figure 3 present the results of an event-study estimation with respect to the timing of the series release.¹⁵ This evidence suggests the absence of pre-trends in terms of interest in *Inspector Montalbano* or in the main municipality where the series was set. Conversely, Figure 3 also shows that people actively searched for keywords related to the series and its location after its release.

Finally, Section 5.2.1 provides additional evidence on the exogeneity of the timing of international broadcasts of *Inspector Montalbano* by presenting event studies on tourism inflows and

¹⁴Residualizing with respect to country fixed effects is necessary to normalize the series, as each Google Trend volume shows changes in Google searches over time within a given country. Since each series does not provide an absolute value for the number of Google searches in a given country-month, it is not possible to compare raw series interest across countries. By taking the residual with respect to country fixed effects, we obtain an index that is more comparable across countries. Year-month fixed effects are also included to control for seasonality.

¹⁵Countries where *Inspector Montalbano* was released before the start of the Google Trends data (i.e., Finland, Greece, France, Sweden and Germany) are excluded from the analysis, since we cannot assess the existence of potential pre-trends for these countries (Miller, 2017). Instead, never-treated countries are included in the analysis, though the pattern does not change when the latter countries are excluded. Results are available upon request.

expenditure, which show the absence of pre-trends.

3 Data

We leverage rich survey data on international tourism in Italy spanning the period from 1997 to 2017. The survey—conducted by the Bank of Italy—is based on interviews of non-resident travelers at the Italian borders (i.e., road and rail crossings, international ports and airports).¹⁶ The survey contains information on the country of origin of each tourist, the municipality visited, the date of each visit and the money spent in each location. In order to have a more homogeneous dataset in terms of tourist inflows, we focus on international tourists coming only from countries belonging to the European Union in 2017 (i.e., at the end of the sample period). Accordingly, we have a sample covering 27 countries of whom around 1/3 is exposed to *Montalbano* by 2017. We end up with a balanced sample of 2,653,560 observations on the number of tourists and on tourist expenditure at the country-municipality-year-month level over the period 1997-2017. As a robustness exercise, we complement this baseline dataset with information on tourism flows coming from the universe of reviews posted on a travel-review website over the period 2002-2017. Specifically, we collected tourist data (i.e., country of origin, date of stay, place visited, etc.) from reviews of 4,670 Sicilian accommodations registered on *Tripadvisor*, the world-leading website of travel-related reviews. We simply count the number of reviews per country of origin posted on the accommodations located in each Sicilian municipality, in each available year-month. As *Tripadvisor* reviews in Italy start in 2002, we build a balanced sample of 2,021,760 observations at the country-municipality-year-month level over the period 2002-2017.¹⁷

In order to implement the analysis on economic development, we first gather data on the supply

¹⁶ "The survey is conducted with the primary objective of compiling the "Travel" item (which includes goods and services purchased by individuals in countries where they are not residents, in relation to travels in those countries) and the "International passenger transport" item of the Italian Balance of payments, in line with the methodological conventions established by the sixth Manual of the IMF." (Bank of Italy, 2019). Bank of Italy adopts a systemic sampling method with fixed n for surveying tourists, and it provides estimated expanded variables on the total number of tourists and on tourist expenditure by country of origin, municipality visited and time of the visit. Additional details are provided in the methodological notes by the Bank of Italy (Bank of Italy, 2017).

¹⁷Beside the limited time period and the absence of information on tourist expenditure, one additional concern regarding *Tripadvisor* data is that they potentially underestimate the tourist inflow for two main reasons: a) they rely exclusively on individuals using the *Tripadvisor*'s services, both tourists and accommodations; b) people who write reviews may travel with other tourists, and we are not able to identify the exact number of people traveling and all their nationalities. For these reasons and given the more limited time span covered by these data, and the absence of information on tourist expenditure, we limit the use of *Tripadvisor* data as a robustness test on our baseline results using the Bank of Italy survey data.

of touristic services at the municipality-year level. Specifically, we exploit data on the number of firms and workers in the tourism sectors (available for 2001 and 2004-2017 from ISTAT).¹⁸ Most importantly, we collect data on total personal income and number of taxpayers at the municipality-year level. These data are provided by the Italian Ministry of Economy and Finance and span the period 2000-2017. As the data on economic outcomes at municipal level are available on a yearly base, we sum tourist expenditure of all countries in our sample at the municipality-year level in order to have a yearly measure of total tourist expenditure for each Sicilian municipality. Accordingly, when we merge data on total tourist expenditure at the municipal level with municipal income to obtain a balanced panel of 390 municipalities over 18 years (2000-2017), ending-up with 7,020 municipality-year observations.¹⁹

4 Empirical strategy

As explained in Section 3, our proxies for tourism—number of tourists and tourist expenditure vary along three dimensions: municipalities, countries, and time. Accordingly, we can estimate a so called triple difference (TD) exploiting the difference between treated and control municipalities and the difference between treated and control countries before and after the series release (Gruber, 1994).²⁰

In particular, we construct a variable measuring the intention to treat with respect to the *Montalbano* series release as the interaction of two indicator variables. Specifically:

$$Montalbano_{c,m,t} = MontalbanoCT_{c,t} \cdot MontalbanoM_m \tag{1}$$

where subscripts c, m, and t stand for country, municipality, and time (year-month), respectively. $MontalbanoCT_{c,t}$ is a dummy that varies at the country-time level and identifies a treated country: it takes a value of one for countries exposed to Montalbano after the time of release onward. Formally,

$$MontalbanoCT_{c,t} = MontalbanoC_c \cdot \mathbb{1}_{t > r_c}$$

 $^{^{18}}$ The 2001 data on the number of firms and tourists in the touristic sectors are gathered from the 2001 census. The data for the period 2004-2017 come from the Italian register of active firms.

¹⁹Similarly, when we merge total tourist expenditure with data on the total number of firms and workers in the tourism sector we end-up with a balanced sample of 390 municipalities over 15 years (census year 2001 and 2004-2017), hence obtaining 5,850 municipality-year observations.

²⁰Frequently, this strategy is also referenced as *difference-in-difference-in-differences* (DDD or DDDs). For simplicity, we adopt the triple difference (TD) terminology as also suggested by Olden and Møen (2020).

where $MontalbanoC_c$ is a dummy variable that varies at the country level, it takes a value of one if a country is ever treated (i.e., if it is ever exposed to the series over our sample period 1997-2017). r_c represents the start of the treatment period in country c, that is the year-month of the *Montalbano* release in country c (normalized to ∞ for countries never treated, i.e., for $MontalbanoC_c = 0$).²¹ $MontalbanoM_m$ is a dummy variable that varies at the municipality level, it simply identifies treated municipalities (i.e., those where the series was staged). That is,

$$Montalbano M_m = \mathbb{1}\{Montalbano Location \subset Municipality_m\}$$

Accordingly, we estimate the following equation:

$$Y_{c,m,t} = \alpha_{cm} + \alpha_{ct} + \alpha_{mt} + \beta^{TD} Montal bano_{c,m,t} + \varepsilon_{c,m,t}$$
⁽²⁾

where $Y_{c,m,t}$ represents the total number of tourists (or tourist expenditure) from country c, in a Sicilian municipality m, and in the year-month t. α -terms capture country-municipality, countryby-time, and municipality-by-time fixed effects, respectively. In particular, the inclusion of α_{cm} allows us to control for any idiosyncratic component at the country-municipality level (e.g., historical tourism or migration links between a municipality and a country). α_{ct} and α_{mt} control non-parametrically for heterogeneous trends in tourism flow dynamics across countries and municipalities, respectively. We adopt a two-way clustering of standard errors at the country and local area (LA) levels (Cameron *et al.*, 2011; Cameron and Miller, 2015).²² Specifically, LAs are clusters of neighbor municipalities officially determined by the Italian National Statistical Institute (IS-TAT) and represent a conservative choice under the assumption that—considering the small size of municipalities in our sample—unobserved components of the outcomes are correlated across units within clusters (Abadie *et al.*, 2017).²³

²¹We define a corresponding starting-treatment year when considering yearly outcomes (e.g., when looking at the impact of yearly total tourist expenditure on municipal income). The starting-treatment year is defined as the year of the first summer holiday season after the release of the series in country c, to account for the heterogeneity of the release-month with respect to such tourism season. That is, it is equal to the release year if the month of release in country c is before September. Instead, it is equal to the year after the release if the month of release in country c is after August. Results are robust to alternative specifications of the treatment year and are available upon request.

 $^{^{22}}$ As a robustness check, we also present estimates for our baseline results adopting the clustering method proposed by Conley (1999), with standard errors corrected for spatial correlation among municipalities that fall within 50 kilometers from each other, and for serial correlation up to 6 months (Table 2). Additionally, in the spirit of Cameron *et al.* (2011), we have considered also a three-way clustering including time (year-month) as a third dimension. Results are also robust under this alternative specification and they are available upon request.

 $^{^{23}}$ There are 73 LAs in Sicily that group 390 municipalities. The mean number of municipalities per LA is 8.6

The TD coefficient of interest, β^{TD} , captures the time change in the average value of the dependent variable for treated countries in *Montalbano* municipalities and then nets out the change in means for control countries in *Montalbano* municipalities, the change in means for treated countries in control municipalities, and the change in means for control countries in control municipalities.²⁴ This specification controls for two kinds of potentially confounding trends: changes in tourist flows—unrelated to the *Montalbano* release—from treated countries across municipalities, and changes in tourist flows in *Montalbano* municipalities possibly due to other municipalities-specific changes that affect the overall tourist flows. Even though the TD estimator can be interpreted as the difference between two difference-in-differences, it does not need two parallel trend assumptions. Rather, the only identifying assumption it requires is that the relative outcome of treated and control countries in Montalbano municipalities trends similarly to the relative outcome of treated and control countries in the control municipalities, in the absence of treatment (Olden and Møen, 2020). Figure 4 provides direct evidence in support of this parallel trend assumption. The graph shows estimated leads and lags coefficients of the difference in the mean number of tourists (left panel) and in the mean tourist expenditure (right panel) in *Montalbano* municipalities (solid line) and in control municipalities (dashed line) with respect to the timing of the *Montalbano* release. Put it differently, the curves compare pre-treatment trends of the two difference-in-differences that enter in the TD estimator. The Figure shows that in the pre-treatment period relative outcomes of treated and control countries have the same trend in both *Montalbano* and control municipalities.²⁵

For what concerns the impact of tourism on economic development, we build a panel dataset at the municipality-year level and we estimate the following model equation:

$$Y_{m,t} = \alpha_m + \alpha_t + \theta_1 T E_{m,t} + \varepsilon_{m,t} \tag{3}$$

where $Y_{m,t}$ measures our outcomes of interest: namely, total personal income, number of taxpayers, and the number of firms and workers in the tourism sector. α_m and α_t represent municipality and year fixed effects, respectively. The variable $TE_{m,t}$ measures the total tourist expenditure of all

with a median equal to 7.

²⁴Formally, $\beta^{TD} = \left[(\bar{Y}_{C,M,2} - \bar{Y}_{C,M,1}) - (\bar{Y}_{c,M,2} - \bar{Y}_{c,M,1}) \right] - \left[(\bar{Y}_{C,m,2} - \bar{Y}_{C,m,1}) - (\bar{Y}_{c,m,2} - \bar{Y}_{c,m,1}) \right]$ where the subscripts *C* and *M* identify treated countries and municipalities, respectively. *c* and *m* stand for control countries and municipalities, and subscripts 1 and 2 simply refer to pre- and post-series release periods.

²⁵Figure A.1 in the Appendix reports the same graph showing also 95% confidence intervals. As expected, the confidence intervals of the leads and lags effects for control municipalities are quite small since the difference in tourism flows from treated and control countries is rather constant over time in those municipalities.

the countries in our sample, in municipality m, and in year t. It is simply the country-sum of the tourist expenditure at the municipality-year level as described in section $3.^{26}$ Yet, total tourist expenditure at the municipality-year level might potentially suffer from endogeneity, for example in terms of reverse causality (e.g., richer municipalities may attract more tourists or invest more in tourism advertisements/services). This would lead to a biased OLS estimation of the equation (3). For this reason, in order to infer the casual relation of interest, we instrument the total tourist expenditure in a municipality-year by exploiting the interaction between a dummy identifying *Montalbano* municipalities and the time-varying share of countries exposed to *Montalbano*. That is, we construct our instrument as follows:

$$ShareExposed_{m,t} = MontalbanoM_m \cdot \left[\frac{\sum_{c} MontalbanoCT_{c,t}}{\sum_{c} \mathbb{1}_{c}}\right]_t$$
(4)

where $\sum_{c} MontalbanoCT_{c,t}$ is the number of countries exposed to the release of *Montalbano* in year $t.^{27} \sum_{c} \mathbb{1}_{c}$ is simply equal to the total number of countries in our sample (27). The exclusion restriction is that changes in the share of countries exposed to *Montalbano* differentially affect *Montalbano* municipalities vs. other Sicilian municipalities only through its effect on total tourist expenditure.

5 The Effect of Media on Tourism

Table 2 shows our baseline results on the effect of being exposed to the *Montalbano* TV series on our two key outcome variables: number of tourists (Columns 1-2) and tourist expenditure (Columns 3-4). Columns 1 and 3 provide estimates with two-way clustering of standard errors at the country and local area (LA) level. Columns 2 and 4 provide estimates when clustering standard errors as suggested by Conley (1999), that is correcting for spatial correlation among municipalities that fall within 50 kilometers from each other, and for serial correlation up to 6 months. Results show a positive and significant impact of the *Montalbano* series on both tourism inflows and tourist expenditure. In terms of magnitudes, we compare our baseline coefficients

 $^{^{26}}$ For what concerns the analysis on the impact of total tourist expenditure on municipal outcomes, since we lose the country dimension we simply cluster standard errors at the local area level. Also in this case, we estimate our model by using a two-way clustering including time (year) as a second dimension (Cameron *et al.*, 2011), results are robust to this alternative specification and they are available upon request.

²⁷Table 8 provides robustness results when considering alternative instrumental variables that differently weight the share of treated countries (e.g., weighting the share of treated countries by the distance from Ragusa, by country population, by country GDP).

with the average outcome of tourists from treated countries in control municipalities (i.e., tourists from countries exposed to the series release not visiting *Montalbano* municipalities, see Table A.1). Accordingly, the implied *media multiplier* (Besley *et al.*, 2020) is such that the series release led to a fourfold increase in the number of tourists and boosted tourist expenditure by a factor of 2.5.²⁸ In the following subsections, we discuss the results of a battery of robustness checks and validity tests.

5.1 Robustness

Table 3 provides several robustness checks. Panel A reports the estimated effect considering the number of tourists as a dependent variable. Panel B presents the corresponding estimates when considering tourist expenditure as a dependent variable. Column 1 shows that our results are robust to considering an alternative treatment variable defined as the interaction between the Montalbano municipality dummy and a continuous variable keeping track of the number of years between the series release and year of the travel. This specification accounts for the fact that some countries are treated for a longer period than others and, therefore, it aims to capture the "intensive margin" of exposure to entertainment media. Moreover, we provide evidence that our results are robust to focusing on alternative data samples at the country or municipality level. In Column 2, we only consider countries that were already belonging to the European Union at the time of the first international release of *Montalbano*. In Column 3, we exclude provincial capitals and, as expected, the coefficients are lower with respect to our baseline specification, as provincial capitals tend to attract the bulk of tourists. In Column 4, we focus on the sub-period 1997-2012. This is relevant as in 2013 a new airport (Comiso) opened in the South-East of Sicily in addition to the existing South-East hub of Catania.²⁹ The estimates remain rather similar to our baseline effects both in terms of magnitudes and significance and both with respect to the number of tourists and tourist expenditure. As the number of tourists is a count variable, in Column 5 of Panel A we estimate a fixed effect Poisson model (Correia et al., 2019) as a robustness check

²⁸Our *media multiplier* effect is almost three times larger than the one found in Besley *et al.* (2020). Such different magnitudes might be potentially driven by several differences of the two contexts under analysis, e.g., direction of the media shock, tourists' prior beliefs on the destination locations and persistence of the shocks.

²⁹While the opening of this airport is likely to have impacted tourism (David and Saporito, 2018), it is worth remarking that all our countries are treated before the opening of the airport. Accordingly, the presence of a new flight connection from treated countries might be by itself seen a possible endogenous effect induced by the release of the TV series. On the other hand, the opening of a flight connection from control countries works against our baseline effect and would lead to a downward bias in our estimates.

of our baseline results.³⁰ Results remain highly statistically significant coefficient. The coefficient of the Poisson specification must be interpreted as a log variable. Hence, it indicates that the release of *Montalbano* in a country leads to an increase by around 65% in tourism inflows from that country in the *Montalbano* municipalities. Column 6 of Panel A presents the results of our baseline specification on the total number of tourists when using observational data from *Tripadvisor* (rather than survey data from Bank of Italy). Notice that in this case the time span is 2002-2017. Results from *Tripadvisor* data are also highly significant with greater implied magnitudes relative to the baseline specification. Specifically, looking at the average outcome of tourists from treated countries in control municipalities in the *Tripadvisor* dataset (see Table A.1), the total number of tourists increases by almost 6 times for the effect of the TV series release. This provides both mutual cross-validation for our two independent sources of data (Bank of Italy and *Tripadvisor*) and it offers a robustness check of our baseline results. Finally, Column 7 of Panel A shows that results are robust when considering the number of nights (rather than the number of tourists) as an alternative tourism outcome.³¹

5.2 Validity Tests

As explained in Section 4, our triple difference empirical strategy only requires the relative difference in tourism outcomes between treated and control countries to exhibit parallel trends in *Montalbano* and control municipalities in the pre-treatment period (Olden and Møen, 2020). Nevertheless, this section presents the results of an event study design showing the absence of pretrends in our main outcomes of interest. Furthermore, we present random inference tests in the spirit of Young (2018) and Dell and Olken (2020). Finally, to address possible concerns related to the staggered nature of our treatment (Goodman-Bacon, 2018; Callaway and Sant'Anna, 2020; De Chaisemartin and d'Haultfoeuille, 2020; Athey and Imbens, 2021) we present estimates when considering one-cohort of treated countries at the time using only never-treated units as controls.

³⁰The number of observations is net of those separated by fixed effects (Correia *et al.*, 2019).

³¹Results (available upon request) are also robust to aggregate outcomes at the year-municipality level rather than at the year-month-municipality level. See also Figure 6 for event-study estimates on the number of tourists and tourist expenditure at the year-municipality level.

5.2.1Event-study

Figure 6 plots the coefficients from the following event-study specification:

$$Y_{c,m,t} = \alpha_{cm} + \alpha_{ct} + \alpha_{mt} + \sum_{j=-\tau}^{\tau} \beta_j Montalbano^j_{c,m,t} + \varepsilon_{c,m,t}$$
(5)

where Y represents our outcome of interest: namely, the number of tourists or tourist expenditure. The α -terms stand for country-municipality, country-by-time, and municipality-by-time fixed effects, respectively. $Montalbano_{c,m,t}^{j}$ are leads and lags of the variable $Montalbano_{c,m,t}$ defined in Equation (4), i.e., with respect to the release year in country c. Specifically,

$$Montalbano_{c.m.t}^{j} = \cdot MontalbanoM_m \cdot MontalbanoC_c \cdot \mathbb{1}_{\{t=r_c+j\}}$$

where r_c , in this case, represents the year in which country c experienced the Montalbano release.³² According to the time-span of our sample, we fix $\tau = 4$, we normalize $\beta_{-1} = 0$, and we use a twoway clustering of standard errors as described in Section 4.³³

Each point in the graph therefore represents the average difference in the dependent variable relative to the treated country-municipality tuple one year before the series release, after controlling for all fixed effects. Overall, Figure 6 highlights the absence of positive pre-trends in the number of tourists or tourist expenditure. If anything, the figure shows a *negative* (albeit small) pre-trend before the ty-series release and it provides evidence of significantly positive effects in the years following the series release. This evidence also complements the analysis, presented in Section 2.2, on the event-study exploiting Google Trends data on searches on words related to the TV series itself ("Montalbano") or location ("Ragusa").

We can speculate that lagged effect is due to some time needed for the touristic supply to adjust. Moreover, the drop at t+3 might be due to some unobserved patterns in re-broadcasting of the series. On the first part: it is true that coefficients are statistically significant and positive before the release. Yet, if anything there is a decreasing trend before/leading to the release and an increasing trend after

All in all, the evidence from these event studies suggests the absence of pre-trends in terms

 $^{^{32}}$ We look at yearly aggregates (rather than year-month outcomes) to provide a clearer picture of the leads and lags effects. Event-study estimates at the monthly level provide similar insights and they are available upon request.

 $^{^{33}}$ We follow McCrary (2008) and bind up end-points. Results are similar for alternative windows.

of interest in the "Inspector Montalbano" series or in the main municipality where the series was staged and, correspondingly, in tourism outcomes toward such locations. At the same time, after the release of the TV series, people actively searched for keywords related to the series and linked locations and, then, visited such locations.

5.2.2 Random Inference Tests

We construct 1,000 counterfactual distributions of our treatment by randomly assigning a treatment to each country-municipality-year-month combination. In particular, we construct counterfactual distributions of the treatment by: a) randomizing the timing of the treatment (i.e., keeping the actual treated country-municipality pairs and randomizing the time of the release of the TV-series); b) randomizing the treated municipalities (i.e., keeping the actual treated country at time t and randomizing the treated municipalities);³⁴ c) Randomizing both over time and over municipalities (i.e., combining a) and b)). Figures 7 and 8 show—for the number of tourists and tourist expenditure, respectively—the distribution of counterfactual estimates over these three dimensions, and compares it to the coefficient from the actual treatment (red vertical line). The distribution of estimated placebo counterfactual effects implies a p-value of 0.000, where the pvalue measures the fraction of the absolute value counterfactual placebo coefficients to the right of the value of the actual coefficient.

5.2.3 Staggered Treatment

To address possible concerns related to the staggered nature of our treatment (Goodman-Bacon, 2018; Callaway and Sant'Anna, 2020; De Chaisemartin and d'Haultfoeuille, 2020; Athey and Imbens, 2021) we present estimates when considering one-cohort of treated countries and at a time, using only never-treated units as controls. In particular, each cohort of estimation includes all control countries (i.e., those never exposed to the series) and treated countries in which the series has been released in the same year. Using information on the TV series release year across countries, we define five cohorts of treated according to whether the *Montalbano* release is in the same year. Given the aggregation by cohorts, as different countries within the same cohort are exposed to the *Montalbano* series in different months, we also aggregate data at the yearly level. Table A.3

³⁴In particular, the randomization process randomly picks four municipalities at a time to be selected as the treated ones, in line with the number of the actual *Montalbano* municipalities.

in the Appendix shows that even in the absence of staggered treatment effects our TD coefficient is positive and statistically significant.

6 Spillover Effects

In this section, we analyze whether the exposure to the TV series created positive spillovers in other Sicilian municipalities. Or, rather, the positive effect on tourism flows to the *Montalbano* municipalities came at the expense of other municipalities (i.e., crowding-out tourists from non-treated municipalities). To answer this question, we focus on a control group constituted by all municipalities in Southern Italy with the exclusion of Sicily. That is, we compare the effect of the exposure to *Montalbano* on tourist flows from country c on all Sicilian municipalities (or subsets of them) versus municipalities in other southern regions. More explicitly, we use the same empirical specification of our baseline regression. Yet, we consider as treated different subsets of municipalities in Sicily (Figure 5) and look at the impact of the TV series on those municipalities compared with other municipalities in the Southern Italy.

Table 4 reports the results. Panel A reports the estimated effect considering the number of tourists as a dependent variable. Panel B presents the corresponding estimates when considering tourist expenditure as a dependent variable. Column 1 reports positive estimates for all municipalities in Sicily (excluding the actual *Montalbano* municipalities from the sample), which are marginally significant for tourist expenditure while imprecisely estimated for the number of tourists. All in all, Column 1 suggests that the gain in tourist inflows experienced by the *Montalbano* municipalities did not come at the expense of other municipalities. Rather, on average, other areas of Sicily benefited from positive spillovers. Consistent with the mechanism, the larger and/or more significant effects are present for municipalities close to the *Montalbano* ones: municipalities in the province of Catania (Column 9) or Siracusa (Column 10). Table 4 confirms that the larger coefficients are indeed the ones obtained considering in the treated group only the municipalities where the series was actually staged (Column 10). At the same time, the results of Column 10 provide evidence on our baseline estimates also being robust when considering a larger control group of municipalities, i.e., looking at the impact of the international release of the tv-series on *Montalbano* municipalities with respect to municipalities in all the other southern Italian regions.

While the Montalbano tv-series did not crowd out tourism in other municipalities in Sicily, it

might be possible that it crowded out other tourism localities elsewhere. Figure A.2 provides event study estimates on aggregate tourist expenditures at the yearly level in each of the other southern Italian regions with respect to the timing of the tv-series release. Such evidence does not seem to suggest the presence of negative spillovers.³⁵

7 The Effect of Tourism on Economic Development

In order to estimate the effect of tourism on economic development by adopting the instrumental variable approach described in section 4, we build a balanced sample at the municipality-year level. In particular, as already mentioned in section 3, we sum total tourist expenditure of all country-month pairs in each municipality for each year in the period 2000-2017. In this way, we build a yearly measure of tourist expenditure (from all countries of our sample) at the municipality-year level.

7.1 Tourism expenditure and the supply of touristic services

As a first step, we analyze the effect of total tourism expenditure (instrumented by Equation (4)) on the supply of touristic services at the municipal level measured by the total number of firms and workers operating in touristic sectors in a municipality-year. Table 5 presents the results. Columns 1 and 2 provide evidence that total tourist expenditure has a positive and significant impact both on the number of firms and number of workers in the tourism sector. In terms of magnitudes, the estimates suggest that a 10% increase in total tourist's expenditure translates into a 11.5% increase in the number of firms and a 10.1% increase in the number of workers in the tourism sector. Accordingly, Table 5 provides evidence of a positive, statistically significant and sizable impact of international tourism expenditure on the equilibrium supply of tourist services.³⁶

³⁵Going beyond Italy, our setting does not allow us to assess the impact of local tourism promotion on the global demand for tourism. Ultimately, the overall effect of an advertising-driven increase in tourism in a given location is likely to depend on the elasticity of the global touristic demand with respect to advertising. If the global demand for tourism were to be perfectly inelastic, promoting tourism localities would be a zero-sum game. Yet, the steady increase in international tourism since WWII (UNWTO, 2019), suggests that advertising a given tourist location does not necessarily hinder the overall tourism flows in other destinations.

³⁶The estimates are robust to controlling for linear or quadratic time-trends at the province or local area level. The results are available upon request.

7.2 Tourism and Municipal Income

Before presenting our main estimates looking at the effect of total tourism expenditure on total municipal income, we provide descriptive evidence on the reduced-form relationship between the international release of the tv-series and municipal income in *Montalbano* and control municipalities. For each year between 2001-2017, Figure 9 shows the difference in means of total municipal income with respect to the baseline year 2000, in *Montalbano* and control municipalities, after controlling for municipality fixed effects.³⁷ The vertical lines represent the sequential release dates in different countries. Thus, the graph provides *prima facie* evidence that the international release of the *Montalbano* tv-series was followed by an increase in municipal income in *Montalbano* municipalities while in other municipalities the total income remained rather flat.

We now turn to our main estimates looking at the effect of total tourism expenditure (instrumented by Equation (4)) on total municipal income. Table 6 shows that an increase of one thousand euros in total tourist expenditure result in an increase of total municipal income by 0.02% (Column 1). Put differently, by relating these quantities with the sample averages on tourist expenditure and total income, we can derive an implied *tourism multiplier* effect such that a 10% increase in total tourist expenditure leads to an increase in total municipal income by 4.7%. Notably, such multiplier is in line with the estimates by Faber and Gaubert (2019) who find that a 10% increase in local hotel revenues leads to a 4% percent increase in nominal municipality GDP in Mexican municipalities. Columns 2-6 shows the total effect decomposed by income brackets. The estimates show a positive and significant effect for all income groups above 16 thousands euros and a negative one below. This suggests that the increase in total tourist expenditure created a shift upward of the income distribution.³⁸

As the share of countries exposed to *Montalbano* is (weakly) monotonically increasing over 37 More formally, we estimate the following equation:

$$Y_{m,t} = \sum_{k=2001}^{2017} \left[\beta_1^k D_k + \beta_2^k D_k \cdot MontalbanoM_m \right] + \delta_m + \varepsilon_{m,t}$$

where $Y_{m,t}$ is the total municipal income divided by 100,000. The variables D_k are dummies for each year k from 2001 to 2017 (since year 2000 is set to be the reference group). Finally, *MontalbanoM* is the dummy that identifies *Montalbano* municipalities, and δ_m are municipality fixed effects. Accordingly, normalizing to 0 the coefficients in year 2000, for each year k from 2001 to 2017, β_1^k is the corresponding estimated coefficient for control municipalities, whereas $\beta_1^k + \beta_2^k$ is the plotted coefficient for *Montalbano* municipalities, for each year k from 2001 to 2017.

³⁸Table A.4 in the appendix provides the OLS estimates. The OLS downward bias suggests that municipalities with a higher income tend to experience a lower impact of total tourist expenditure on overall income. This might be due to the fact that, on average, in richer municipalities a lower share of municipal income relates to tourism.

time, differences in local trends might potentially confound our estimates. Table 7 addresses this issue by providing estimates of the robustness of the impact of tourism on municipal income when controlling for linear (Column 1) or quadratic (Column 2) time-trends at the province level or at the local area level (Columns 3 and 4). Including local time-trends allows—among others—to control for any unobservable change that occur at the province or even at the local area level in a given year. For example, it allows to control for yearly variation in domestic tourism towards a given area or any other yearly variation in local economic activities. While the point estimates are lower with respect to the baseline ones presented above (suggesting that local trends might indeed play some role), they remain positive, statistically significant and sizable. In addition, the first stage becomes even stronger. Table 8 provides further robustness checks. In particular, it shows that results are consistent when focusing only on the subset of countries that were already in the European Union in 2000, i.e., before the first international release of the Montalbano series. Similarly, results are robust when excluding provincial capital, or when considering only the period before the opening of the Comiso airport in the South-East of Sicily, i.e., before 2013. Finally, considering alternative instrumental variables that differently weight the share of treated countries (e.g., weighting the share of treated countries by the distance from Ragusa, by country population, by country GDP) leads to very similar estimates both in terms of significance and magnitudes.

In order to better understand the mechanism and the drivers, Table 9 provides estimates by decomposing the overall effect of total tourist expenditure on municipal income in the extensive and intensive margin. Specifically, Panel A (Columns 1-6) looks at the impact of total TE on the number of taxpayers. The estimates in Column 1 suggest that a 10% increase in total TE leads to an increase in municipal taxpayers by 0.64%. Columns 2-6 present the effects by income brackets, showing that—consistent with the results of Table 6—the positive effect on the extensive margin is driven by an increase in the number of taxpayers with an income above 16,000 euros while, if anything, it leads to a drop in the number of taxpayers in the lowest income group (below 16,000 euros). Panel B shows that total tourist expenditure also have a positive effect on the municipal income per-taxpayer. The estimates in Column 7 suggest that a 10% increase in total TE leads to a 0.05% increase in municipal income per-taxpayer. Columns 8-12 show the presence of some negative effects on the upper-brackets due to a larger increase in the denominator (i.e., in the overall number of taxpayers in those income classes). All in all, the results provide evidence that

tourist expenditure might be a significant driver of economic development.³⁹

8 Concluding Remarks

We provide evidence of the impact of entertainment media on tourism and, in turn, the overall impact of tourism on municipal income. This is done by leveraging rich data on international tourism and exploiting the staggered international release of *Inspector Montalbano*, an Italian TV series set in southeastern Sicily. We document that exposure to the series drew tourists toward the filming locations, with positive spillovers also observed in nearby areas. Furthermore, we provide evidence of a positive and significant impact of tourism on economic development using the interaction between a dummy variable identifying *Montalbano* municipalities and the timevarying share of countries in which the series had been released, as an instrument for total tourist expenditure in a municipality.

In terms of the *media multiplier* (Besley *et al.*, 2020), exposure to *Inspector Montalbano* translates into an increase in tourist expediture of a factor of about 2.5. Our implied *media multiplier* is thus almost three times larger than that found in Besley *et al.* (2020). We speculate that such differences in magnitude might be driven by several factors. First, it could be due to the different direction of the media effect with respect to viewers' beliefs on the desirability of a given destination (positive in our case and negative in Besley *et al.* 2020). Second, in our context, destinations were likely to be completely unknown to viewers prior to the series release, which may have amplified the media effect. Finally, as noted by Besley *et al.* (2020), the persistence of a media shock (e.g., the broadcasting or re-broadcasting of episodes over time) might also result in a larger *media multiplier*. Future research on the impact of different types of media shocks in different contexts may prove helpful in assessing the range and drivers of such a *media multiplier*.

It is also possible to relate the media effects we identify with similar effects reported in the literature on advertising and consumer persuasion (Bertrand *et al.*, 2010; DellaVigna and Gentzkow, 2010; Stephens-Davidowitz *et al.*, 2017; Hartmann and Klapper, 2018; Valletti and Veiga, 2021). In this respect, our estimates suggest that exposure to *Inspector Montalbano* may generate an

³⁹As our main dependent variable is given by the total declared personal income aggregated at the municipalityyear level, our effects might be biased upward or downward if tourism affects the propensity of declaring income to the tax authorities, in a direction or another. The estimated effects might be an upper bound of the true effect of tourism on municipal income if international tourism increases such propensity. Instead, it might represent a lower bound, if part of the tourist expenditure goes into the informal economy.

effect akin to the exposure to three Super Bowl ads (Stephens-Davidowitz *et al.*, 2017).⁴⁰ Or, stated in terms of the elasticity of consumers' choices with respect to viewing time, the impact of *Inspector Montalbano* on tourist inflows could be comparable to looking at an online advertisement for an extra 7 minutes (Valletti and Veiga, 2021). Overall, the larger magnitudes of the effects we observe are likely due to few key differences between entertainment media and traditional ads. In our case, the implicit advertisement of the filming locations is embedded within a well-received entertainment product, while traditional ads are typically considered a nuisance (Anderson and Gabszewicz, 2006). Furthermore, the series features several episodes that last around one hour each. As such, the length and repetition of the message may help to amplify its effect with respect to the one of traditional ads.

Finally, in terms of the *tourism multiplier*—in line with the estimates of Faber and Gaubert (2019) on Mexican municipalities—we show that a 10% increase in total tourist expenditure translates into a 4.7% increase in municipal income. This boost in municipal income also came with a shift of taxpayers from lower to higher income brackets and, as expected, created more jobs and firms in the tourist sector.

Our findings are relevant to inform business-owners and policy-makers on media and tourism multipliers. That is, on the potential returns of investing in intangibles such as promoting tourist destinations, and on the overall impact of tourism on the local economy. Indeed, while identifying the successful features of a media product is clearly beyond the scope of the paper, *Inspector Montalbano* did showcase attractive elements of the local destinations, such as natural and historical landscapes and local cuisine. As such, our results provide a possible economic rationale for granting public subsidies for film productions (European Commission, 2014; KOFIC, 2020).⁴¹ All in all, this paper suggests that both entertainment media and tourism can be useful tools for boosting local economic development.

⁴⁰Admittedly, this is a rather approximate back-of-the-envelope calculation as data on *Inspector Montalbano* ratings across countries and time are not available.

⁴¹In order to access public subsidies under this conditionality clause, public authorities would require a minimum amount of shooting days in a given location.

References

- ABADIE, A., ATHEY, S., IMBENS, G. W. and WOOLDRIDGE, J. (2017). When should you adjust standard errors for clustering? *NBER Working Paper*.
- ACCETTURO, A. and DE BLASIO, G. (2019). Morire di Aiuti: I Fallimenti delle Politiche per il Sud (e come evitarli). IBL libri.
- ALBANESE, G., DE BLASIO, G. and LOCATELLI, A. (2020). Does eu regional policy promote local tfp growth? evidence from the italian mezzogiorno. *Papers in Regional Science*.
- ANDERSON, S. P. and GABSZEWICZ, J. J. (2006). The media and advertising: A tale of two-sided markets. *Handbook of the Economics of Art and Culture*, **1**, 567–614.
- ANELLI, M. and PERI, G. (2017). Does emigration delay political change? Evidence from Italy during the great recession. *Economic Policy*, **32** (91), 551–596.
- AREZKI, R., CHERIF, R. and PIOTROWSKI, J. (2009). Tourism specialization and economic development: Evidence from the unesco world heritage list. *IMF Working Paper 2009/176*.
- ATHEY, S. and IMBENS, G. W. (2021). Design-based analysis in difference-in-differences settings with staggered adoption. *Journal of Econometrics*.
- BALLARINO, G., BRAGA, M., BRATTI, M., CHECCHI, D., FILIPPIN, A., FIORIO, C. V., LEONARDI, M., MESCHI, E. and SCERVINI, F. (2013). GINI Country Report: Growing Inequalities and their Impacts in Italy. Working Paper Amsterdam Institute for Advanced Labour Studies.
- BANDIERA, O. (2003). Land reform, the market for protection, and the origins of the sicilian mafia: Theory and evidence. Journal of Law, Economics, & Organization, 19 (1), 218–244.
- BANK OF ITALY (2017). Survey on International Tourism: Methodogical Notes. https: //www.bancaditalia.it/pubblicazioni/metodi-e-fonti-note/metodi-note-2017/ tourism_note_met-en.pdf?language_id=1, [Online; accessed 01-March-2020].
- BANK OF ITALY (2019). Survey on International Tourism. https://www.bancaditalia.it/ statistiche/tematiche/rapporti-estero/turismo-internazionale/index.html?com. dotmarketing.htmlpage.language=1, [Online; accessed 01-March-2020].
- BARONE, G. and MOCETTI, S. (2014). Natural disasters, growth and institutions: A tale of two earthquakes. *Journal of Urban Economics*, 84, 52–66.
- BARTOLUCCI, C., VILLOSIO, C. and WAGNER, M. (2018). Who migrates and why? evidence from italian administrative data. *Journal of Labor Economics*, **36** (2), 551–588.
- BBC NEWS (2015). Stirling Screening Marks Braveheart Film Anniversary. https://www.bbc.com/news/uk-scotland-tayside-central-34127673, [Online; accessed 01-July-2020].
- BECKER, S. O., EGGER, P. H. and VON EHRLICH, M. (2013). Absorptive capacity and the growth and investment effects of regional transfers: A regression discontinuity design with heterogeneous treatment effects. *American Economic Journal: Economic Policy*, 5 (4), 29–77.

- —, and (2018). Effects of eu regional policy: 1989-2013. Regional Science and Urban Economics, 69, 143 152.
- -, ICHINO, A. and PERI, G. (2004). How large is the "brain drain" from italy? *Giornale degli* economisti e annali di economia, pp. 1–32.
- BEETON, S. (2016). Film-Induced Tourism. Channel view publications.
- BERTRAND, M., KARLAN, D., MULLAINATHAN, S., SHAFIR, E. and ZINMAN, J. (2010). What's advertising content worth? evidence from a consumer credit marketing field experiment. *The Quarterly Journal of Economics*, **125** (1), 263–306.
- BESLEY, T. J., FETZER, T. and MUELLER, H. F. (2020). How Big is the Media Multiplier? Evidence from Dyadic News Data. *Unpublished manuscript*.
- BORDIGNON, M., CERNIGLIA, F. and REVELLI, F. (2003). In search of yardstick competition: a spatial analysis of italian municipality property tax setting. *Journal of Urban Economics*, **54** (2), 199–217.
- BUONANNO, P., DURANTE, R., PRAROLO, G. and VANIN, P. (2015). Poor institutions, rich mines: Resource curse in the origins of the sicilian mafia. *The Economic Journal*, **125** (586), F175–F202.
- BUSBY, G. and KLUG, J. (2001). Movie-induced Tourism: The Challenge of Measurement and Other Issues. *Journal of Vacation Marketing*, **7** (4), 316–332.
- BUSINESS INSIDER (2019). One Of The 7 Wonders Of The World Is A 10,000-Year-Old City Hidden In The Desert. https://www.businessinsider.com/ petra-jordan-7-wonders-of-the-world-indiana-jones-2018-7, [Online; accessed 01-July-2020].
- CALLAWAY, B. and SANT'ANNA, P. H. (2020). Difference-in-differences with multiple time periods. *Journal of Econometrics*.
- CAMERON, A. C., GELBACH, J. B. and MILLER, D. L. (2011). Robust inference with multiway clustering. *Journal of Business & Economic Statistics*, **29** (2), 238–249.
- and MILLER, D. L. (2015). A practitioner's guide to cluster-robust inference. Journal of Human Resources, 50 (2), 317–372.
- CASELLI, G. (2017). Il Turismo Invisibile. https://www.ucer.camcom.it/comunicazione/ notizie/pdf-2017/turismoinvisibile.pdf, [Online; accessed 01-March-2020].
- CHOWDHURY, A. (1994). Advertising expenditures and the macroeconomy: some new evidence. International Journal of Advertising, 13 (1), 1–14.
- CONLEY, T. G. (1999). Gmm estimation with cross sectional dependence. *Journal of Economet*rics, **92** (1), 1–45.
- CORREIA, S., GUIMARÃES, P. and ZYLKIN, T. (2019). Verifying the existence of maximum likelihood estimates for generalized linear models. arXiv preprint arXiv:1903.01633.

- D'AMURI, F., OTTAVIANO, G. I. and PERI, G. (2010). The labor market impact of immigration in western germany in the 1990s. *European Economic Review*, **54** (4), 550–570.
- DAVID, F. and SAPORITO, G. (2018). The impact of a new airport on international tourism: The case of ragusa (sicily, italy). *Politica economica, Journal of Economic Policy*, (3/2018), 327–344.
- DE CHAISEMARTIN, C. and D'HAULTFOEUILLE, X. (2020). Two-way fixed effects estimators with heterogeneous treatment effects. *American Economic Review*, **110** (9), 2964–96.
- DELL, M. and OLKEN, B. A. (2020). The development effects of the extractive colonial economy: The dutch cultivation system in java. *The Review of Economic Studies*, 87 (1), 164–203.
- DELLAVIGNA, S. and GENTZKOW, M. (2010). Persuasion: Empirical evidence. Annual Review of Economics, 2 (1), 643–669.
- and LA FERRARA, E. (2015). Economic and Social Impacts of the Media. In Handbook of Media Economics, vol. 1, Elsevier, pp. 723–768.
- EILAT, Y. and EINAV, L. (2004). Determinants of international tourism: A three-dimensional panel data analysis. *Applied Economics*, **36** (12), 1315–1327.
- EUROPEAN COMMISSION (2014). Competition Policy Brief. State Aid Rules For Films And Other Audiovisual Works. https://ec.europa.eu/competition/publications/cpb/2014/013_en. pdf, [Online; accessed 01-March-2020].
- EUROSTAT (2021). Crimes Recorded By The Police By NUTS 3 Regions. https://ec.europa.eu/ eurostat/databrowser/product/view/CRIM_GEN_REG?lang=en, [Online; accessed 01-March-2021].
- FABER, B. and GAUBERT, C. (2019). Tourism and Economic Development: Evidence from Mexico's Coastline. American Economic Review, 109 (6), 2245–93.
- FEDERICO, G., NUVOLARI, A. and VASTA, M. (2019). The origins of the italian regional divide: Evidence from real wages, 1861–1913. *The Journal of Economic History*, **79** (1), 63–98.
- FORBES (2012). The Impact (Economic and Otherwise) of Lord of the Rings/The Hobbit on New Zealand. https://www.forbes.com/sites/carolpinchefsky/2012/12/14/ the-impact-economic-and-otherwise-of-lord-of-the-ringsthe-hobbit-on-new-zealand, [Online; accessed 01-July-2020].
- GALLETTA, S. (2017). Law enforcement, municipal budgets and spillover effects: Evidence from a quasi-experiment in italy. *Journal of Urban Economics*, **101**, 90–105.
- GARDINER, B., MARTIN, R., SUNLEY, P. and TYLER, P. (2013). Spatially unbalanced growth in the British economy. *Journal of Economic Geography*, **13** (6), 889–928.
- GOODMAN-BACON, A. (2018). Difference-in-differences with variation in treatment timing. *NBER* Working Paper.

- GROSSMANN, V. (2008). Advertising, in-house r&d, and growth. Oxford Economic Papers, **60** (1), 168–191.
- GRUBER, J. (1994). The incidence of mandated maternity benefits. *The American Economic Review*, pp. 622–641.
- HARTMANN, W. R. and KLAPPER, D. (2018). Super bowl ads. *Marketing Science*, **37** (1), 78–96.
- HEISE, S. and PORZIO, T. (2019). Spatial wage gaps and frictional labor markets. *FRB of New York Staff Report*, (898).
- HINNOSAAR, M., HINNOSAAR, T., KUMMER, M. E. and SLIVKO, O. (forthcoming). Wikipedia matters. Journal of Economics & Management Strategy.
- HUDSON, S. and RITCHIE, J. R. B. (2006a). Film Tourism and Destination Marketing: The Case of Captain Corelli's Mandolin. *Journal of Vacation Marketing*, **12** (3), 256–268.
- and (2006b). Promoting Destinations via Film Tourism: An Empirical Identification of Supporting Marketing Initiatives. *Journal of Travel Research*, 44 (4), 387–396.
- KOFIC (2020). KOFIC Location Incentive for Foreign Audio-visual Works. http://www.koreanfilm.or.kr/eng/coProduction/locIncentive.jsp, [Online; accessed 01-March-2020].
- LINTZ, G. and SCHMUDE, K. (2005). *Germany: Tackling the East-West Divide*, Springer Berlin Heidelberg, pp. 81–110.
- MARTIN, R. (1988). The political economy of britain's north-south divide. Transactions of the Institute of British Geographers, **13** (4), 389–418.
- MCCRARY, J. (2008). Manipulation of the running variable in the regression discontinuity design: A density test. *Journal of Econometrics*, **142** (2), 698–714.
- MCGREGOR, T. and WILLS, S. (2017). Surfing a wave of economic growth. CAMA Working Paper.
- MILLER, C. (2017). The persistent effect of temporary affirmative action. American Economic Journal: Applied Economics, 9 (3), 152–90.
- MOLINARI, B. and TURINO, F. (2018). Advertising and aggregate consumption: A bayesian dsge assessment. *The Economic Journal*, **128** (613), 2106–2130.
- OLDEN, A. and MØEN, J. (2020). The triple difference estimator. NHH Dept. of Business and Management Science Discussion Paper, (2020/1).
- RILEY, R., BAKER, D. and VAN DOREN, C. S. (1998). Movie induced tourism. Annals of Tourism Research, 25 (4), 919–935.
- RILEY, R. W. and VAN DOREN, C. S. (1992). Movies as tourism promotion: A 'pull'factor in a 'push'location. *Tourism Management*, **13** (3), 267–274.

- SELDOM, B. J. and JUNG, C. (1995). The length of the effect of aggregate advertising on aggregate consumption. *Economics Letters*, **48** (2), 207–211.
- SEQUEIRA, T. N. and MAÇÃS NUNES, P. (2008). Does tourism influence economic growth? a dynamic panel data approach. *Applied Economics*, **40** (18), 2431–2441.
- SOLIMAN, D. M. (2011). Exploring the role of film in promoting domestic tourism: A case study of al fayoum, egypt. *Journal of Vacation Marketing*, **17** (3), 225–235.
- STEPHENS-DAVIDOWITZ, S., VARIAN, H. and SMITH, M. D. (2017). Super returns to super bowl ads? *Quantitative Marketing and Economics*, **15** (1), 1–28.
- TAYLOR, L. D. and WEISERBS, D. (1972). Advertising and the aggregate consumption function. The American Economic Review, 62 (4), 642–655.
- TELEGRAPH, T. (2019). The Game Of Thrones Effect How The TV-Series Changed How We Travel. https://www.telegraph.co.uk/travel/destinations/europe/articles/ game-of-thrones-effect-tourism/, [Online; accessed 01-July-2020].
- TSUI, K.-Y. (2007). Forces Shaping China's Interprovincial Inequality. Review of Income and Wealth, 53 (1), 60–92.
- UNWTO (2019). Unwto world tourism barometer and statistical annex, september 2019. UNWTO World Tourism Barometer (English version), 17 (3), 1–40.
- USA TODAY (2013). Breaking Bad Has Been Very Good For Albuquerque. https://eu.usatoday.com/story/travel/destinations/2013/08/11/ breaking-bad-albuquerque-tourism/2636859/, [Online; accessed 01-July-2020].
- VALLETTI, T. M. and VEIGA, A. (2021). Attention, Recall and Purchase: Experimental Evidence on Online News and Advertising. Tech. rep., CESifo Working Paper.
- WORLD BANK (2020). World Bank, International Comparison Program database. https://data. worldbank.org/indicator/NY.GDP.PCAP.PP.CD, [Online; accessed 01-March-2020].
- WORLD TRAVEL TOURISM COUNCIL (2019). Italy. 2019 Annual Research: Key Highlights. https: //www.wttc.org/economic-impact/country-analysis/country-data, [Online; accessed 01-March-2020].
- YOUNG, A. (2018). Channeling fisher: Randomization tests and the statistical insignificance of seemingly significant experimental results. *The Quarterly Journal of Economics*, **134** (2), 557–598.

Figures and Tables



Figure 1: Montalbano Municipalities



Figure 2: Countries exposed to Montalbano

Note: The graph shows the map of EU countries (excluding Italy) in 2017. Darker (blue) areas correspond to countries exposed to the *Montalbano* TV series over our sample period (1997-2017). The number inside the dark areas refer to the year of the TV series release.



Figure 3: Google-trends

Note: Figures refer to G-trends data on the keywords "Montalbano" and "Ragusa" from 2004 to 2017. Left-side figures show the average monthly residual of Google Trends search volumes in origin countries with respect to the country fixed effects (to account for the non-comparability of G-trends hits across countries) and year-month fixed effects (to account for seasonality). The left panel compares the mean (with confidence intervals) of such residual in the period before the TV series release and after. Right-side figures show event-study estimates at the yearly-level with country and year fixed effects.

Figure 4: Tourism Trends



Note: The graph shows the estimated leads and lags coefficients of the difference in the mean number of tourists (left panel) and in the mean tourist expenditure (right panel) in *Montalbano* municipalities (solid line) and in control municipalities (dashed line) with respect to the timing of the *Montalbano* series release.



Figure 5: Spillovers - Treatment Groups

Figure 6: Event Study: Number of Tourists & Tourist Expenditure









Figure 8: Random Inference Test: Tourist Expenditure



Figure 9: International tv-series Release and Municipal income

Note: The graph shows point estimates (with 95% confidence intervals) of the impact on total municipal income (per 100,000 euros) of year dummies and the interaction between year dummies and the Montalbano municipality dummy, after controlling for municipality fixed effects and normalizing to zero the coefficients in year 2000.

| | Randomization Test | Balance Test |
|---|--------------------|--------------------|
| Demo & Geographic Characteristics: | | |
| Resident population | 0.000 | 26424 |
| | (0.000) | (20963) |
| Share of population aged 20-34 y.o. | 0.291 | 0.010 |
| | (0.267) | (0.010) |
| Share of population aged above 65 y.o. | -0.081 | -0.022 |
| ~ | (0.098) | (0.026) |
| Share of population with a high school/university diploma | 0.053 | 0.021 |
| | (0.080) | (0.032) |
| High level of urbanization | -0.011 | -0.096 |
| | (0.017) | (0.148) |
| Totally montaneous territory | 0.009 | 0.227 |
| | (0.010) | (0.251) |
| Partially montaneous territory | 0.002 | 0.038 |
| NT / / | (0.012) | (0.206) |
| Non montaneous territory | -0.014 | -0.264 |
| | (0.012) | (0.221) |
| Economic Characteristics: | | |
| Tot Dersonal Income per capita | 0.000 | 775.08 |
| fot. Personal income per capita | (0.000) | (545, 33) |
| N of production units | (0.000) | (040.00) 1 763 |
| N. of production units | (0.000) | (1.006) |
| N of non profit organizations | (0.000) | 110.0 |
| iv. or non-prone organizations | (0,000) | (73.66) |
| N of employees in non profit organizations | 0.000) | 232.9 |
| 1. of employees in non-prone organizations | (0.000) | (217.5) |
| Tourism Characteristics: | (0.000) | (211.0) |
| N tourists por conito | 0.001 | 0.008 |
| N. tourists per capita | (0.001) | (0.106) |
| Total touvist synanditure non conita | (0.013) | (0.190) |
| iotai tourist experianture per capita | (0.000) | (162.82) |
| N of firms in touristm sectors | (0.000) | (102.82) |
| N. OI IIIIIS III toulistiii sectors | (0.000) | (50.43) |
| N of employees in tourism sectors | | (30.43) 973 74 |
| 11. of employees in fourism sectors | (0,000) | 213.14 (177.48) |
| | (0.000) | (111.40) |
| Observations | 390 | 390 |

Note: All the variables refer to census year 2001. Randomization test is computed regressing the instrument on each variable. Balance test is computed regressing each variable on the instrument. Standard errors in parentheses. The variable "N. tourists per capita" refers to the total number of tourists from European Union countries visiting a municipality in 2001 (source: Bank of Italy), divided by the resident municipal population in 2001. The variable "Tourist expenditure per capita" refers to the total expenditure of tourists from European Union countries in the municipality in 2001 (source: Bank of Italy), divided by the resident municipal population in 2001. *, **, *** indicate statistical significance at 10%, 5%, and 1% level, respectively.

| Table 2: Media & Tourisms | | | | | | | | |
|--|---------------------------|----------------------------|---------------------------|---------------------------|--|--|--|--|
| | (1) | (2) | (3) | (4) | | | | |
| | Number of | Number of | Tourist | Tourist | | | | |
| | Tourists | Tourists | Expenditure | Expenditure | | | | |
| Montalbano mun. \times Country treated | $71.188^{***} \\ (4.365)$ | $71.188^{***} \\ (13.917)$ | 35.312^{***} (2.958) | 35.312^{**} (15.769) | | | | |
| Observations | $2,\!653,\!560$ | $2,\!653,\!560$ | $2,\!653,\!560$ | $2,\!653,\!560$ | | | | |
| Country-Municipality FE | YES | YES | YES | YES | | | | |
| Country-Time FE | YES | YES | YES | YES | | | | |
| Municipality-Time FE | YES | YES | YES | YES | | | | |
| SE clustered at Country & LA | YES | NO | YES | NO | | | | |
| SE Conley | NO | YES | NO | YES | | | | |

Note: Time horizon 1997-2017. Tourist expenditure is reported in thousands of euros. Columns (1) and (3) and (5): standard errors clustered at Country and Local Area levels in parentheses. Columns (2) and (4): Conley standard errors corrected for spatial correlation among municipalities that fall within 50 km from each other, and for serial correlation up to 6 months, in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1

Table 3: Media & Tourisms: Robustness Check

| | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
|---|---|--------------------------------|--|--------------------------------|---|---|--------------------------------|
| | Years since Montalbana | Only FU 2000 | Drop | Pre- | Poisson | Tripadvisor | N. of Nighta |
| | release | Countries | Capitals | 2013 | | Data | Inights |
| | | | | | | | |
| Panel A: Number of Tourists | | | | | | | |
| $\begin{array}{l} \mbox{Montalbano Mun.} \\ \times \mbox{ Years since release} \end{array}$ | 9.567^{***} (0.701) | | | | | | |
| $\begin{array}{l} \mbox{Montalbano Mun.} \\ \times \mbox{ Country Treated} \end{array}$ | | 69.011^{***} (4.371) | 26.196^{***} (1.411) | 61.471^{***} (5.040) | $\begin{array}{c} 0.657^{***} \\ (0.100) \end{array}$ | $\begin{array}{c} 0.702^{***} \\ (0.126) \end{array}$ | 0.702^{***} (0.045) |
| Panel B: Tourist Expenditure | | | | | | | |
| Montalbano Mun. \times Years since release | $\begin{array}{c} 2.061^{***} \\ (0.390) \end{array}$ | | | | | | |
| Montalbano Mun. \times Country Treated | | 38.632^{***} (3.122) | $\begin{array}{c} 13.228^{***} \\ (0.980) \end{array}$ | 33.434^{***} (3.941) | - | - | - |
| Observations Country-Municipality FE Country-Time FE Municipality-Time FE | 2,653,560 YES YES YES | 1,375,920 YES YES YES | 2,592,324 YES YES YES | 2,021,760 YES YES YES | 89,943 YES YES YES | 2,021,760 YES YES YES | 2,653,560 YES YES YES |

Note: Tourist expenditure is reported in thousands of euros. Time horizon: 1997-2017 for Columns 1-2-3, 5 and 7; 1997-2012 for Column 4 and 2002-2017 for Column 6. In Panel A: Column 5 reports the estimate of a Poisson model; Column 6 reports the estimated effect on the Number of Tourists using the *Tripadvisor* dataset; Column 7 reports the estimated effect when considering the number of nights as a dependent variable. Standard errors clustered at Country and Local Area levels in parentheses. *** p<0.01, ** p<0.05, * p<0.1

| 1able 4. | Media & 10011 | unde same | Mer Eneco | 5 | |
|---------------------------------------|------------------|------------------|------------------|------------------|----------------|
| | (1) | (2) | (3) | (4) | (5) |
| Treated Groups: | Sicily | Trapani | Palermo | Messina | Agrigento |
| F. | (no Montalbano | province | province | province | province |
| | municipalities) | province | province | province | province |
| | municipantics) | | | | |
| Panel A: Number of Tourists | | | | | |
| Treated Mun × Country Treated | 5.624 | 6 845 | 10.632 | 4 356 | -8 795 |
| ficated Mail: × Country ficated | (3.412) | (5.497) | (7.911) | (4.126) | (5.214) |
| | (0.412) | (0.451) | (1.511) | (4.120) | (0.214) |
| | | | | | |
| Panel B. Tourist Expenditure | | | | | |
| Treated Mun × Country Treated | 5 971* | 6 502 | 9 621 | 4 043 | -1.860 |
| ficated Mull: × Country ficated | (2.967) | (4.055) | (6.724) | (3.663) | (1.121) |
| | (2.501) | (4.000) | (0.124) | (0.000) | (1.121) |
| | | | | | |
| Observations | 17.363.808 | 14.900.760 | 15,295,392 | 15.472.296 | 15.030.036 |
| | 1,000,000 | 11,000,100 | 10,200,002 | 10,112,200 | 10,000,000 |
| | (6) | (7) | (8) | (9) | (10) |
| Treated Groups: | Caltanissetta | Enna | Catania | Siracusa | Montalbano |
| F. | province | province | province | province | municipalities |
| | province | province | province | province | municipantico |
| Panel A. Number of Tourists | | | | | |
| Tanel A. Rumber of Tourists | | | | | |
| Treated Mun × Country Treated | -2.669 | 1 005 | 10 715* | 21 003* | 77 056*** |
| ficated with × country ficated | (1.674) | (2.265) | (5.923) | $(12\ 173)$ | (1.924) |
| | (1.011) | (2.200) | (0.020) | (12.110) | (1.021) |
| | | | | | |
| Panel B: Tourist Expenditure | | | | | |
| I I I I I I I I I I I I I I I I I I I | | | | | |
| Treated Mun. \times Country Treated | -0.268 | 1.153 | 7.333 | 13.198* | 40.794*** |
| | (1.364) | (1.749) | (4.482) | (7.481) | (1.176) |
| | (| () | () | () | () |
| | | | | | |
| Observations | $14,\!887,\!152$ | $14,\!873,\!544$ | $15,\!132,\!096$ | $14,\!880,\!348$ | 14,764,680 |
| Country-Municipality FE | YES | YES | YES | YES | YES |
| Country-Time FE | YES | YES | YES | YES | YES |
| Municipality-Time FE | YES | YES | YES | YES | YES |

Table 4: Media & Tourisms: Spillover Effects

Note: Time horizon 1997-2017. Tourist expenditure is reported in thousands of euros. Each column is titled with the name of the province or group of municipalities treated. Control group of Italian southern regions: Abruzzo, Basilicata, Calabria, Campania, Molise, Puglia, and Sardegna. Standard errors clustered at Country and Local Area levels in parentheses. *** p<0.01, ** p<0.05, * p<0.1

| 1 | 11 / | |
|-----------------------------------|---|---|
| | (1) | (2) |
| | N. of Firms | N. of Workers |
| Estimation Method: IV | Tourism Sectors | Tourism Sectors |
| Total TE | $\begin{array}{c} 0.0282^{***} \\ (0.0034) \end{array}$ | $\begin{array}{c} 0.0817^{***} \\ (0.0091) \end{array}$ |
| Observations | 5,850 | 5,850 |
| Municipality FE | YES | YES |
| Year FE | YES | YES |
| F-stat of the Excluded Instrument | 53.56 | 53.56 |

Table 5: Tourism Expenditure and he Supply of Touristic Services

Note: Time horizon 2001, and 2004-2017. Data on the number of firms and workers in the tourism sector at the year-municipality level are available from 2004, data for the year 2001 refer to census data. Total TE (Total Tourist Expenditure) is reported in thousands of euros. Standard errors clustered at Local Area level in parentheses. The Table reports instrumental variable estimates. *** p<0.01, ** p<0.05, * p<0.1

| | 0. 100101 | n ana Been | | pinone | | |
|--------------------------------|---------------------------|----------------------|------------------------|---------------------------|------------------------|------------------------|
| | (1) | (2) | (3) | (4) | (5) | (6) |
| | Total | Income | Income | Income | Income | Income |
| Estimation: IV | Income | below $15K$ | 16K-25K | 26K-55K | 56K-75K | above $75K$ |
| Total TE | $25,212^{***}$ (2,750) | $-1,247^{***}$ (165) | $6,735^{***}$ (980) | $14,723^{***}$ (1,569) | $1,626^{***}$ (161) | $3,375^{***}$ (333) |
| Observations | 7,020 | 7,020 | 7,020 | 7,020 | 7,020 | 7,020 |
| Municipality FE | YES | YES | YES | YES | YES | YES |
| Year FE | YES | YES | YES | YES | YES | YES |
| F-stat of the Excl. Instrument | 49.60 | 49.60 | 49.60 | 49.60 | 49.60 | 49.60 |

Table 6: Tourism and Economic Development

Note: Time horizon 2000-2017. Total TE (Total Tourist Expenditure) is reported in thousands of euros. Outcome variables refer to total taxable income in the municipality-year (Column 1) or the total personal income within a given income class (Columns 2-6). Standard errors clustered at Local Area level in parentheses. The Table reports instrumental variable estimates. *** p<0.01, ** p<0.05, * p<0.1

| Estimation Method: IV | (1) | (2) | (3) | (4) | | | | |
|-----------------------------------|----------------|-----------------|----------------|-----------------|--|--|--|--|
| | Provincial | Provincial | Local Area | Local area | | | | |
| | linear trend | quadratic trend | linear trend | quadratic trend | | | | |
| Total TE | $11,988^{***}$ | $11,599^{***}$ | $16,231^{***}$ | $15,255^{***}$ | | | | |
| | (2,596) | (2,298) | (396) | (298) | | | | |
| Observations | 7,020 | 7,020 | 7,020 | 7,020 | | | | |
| Municipality FE | YES | YES | YES | YES | | | | |
| Year FE | YES | YES | YES | YES | | | | |
| F-stat of the Excluded Instrument | 114.4 | 130.2 | 1884 | 2491 | | | | |

Table 7: Tourism and Economic Development - Robustness

Note: Dependent variable: total municipal income. Total TE (Total Tourist Expenditure) is reported in thousands of euros. Columns 1 and 2 provide IV estimates of the impact of total tourism expenditure on total municipal income when controlling for a linear and quadratic time-trends at the province level, respectively. Columns 3 and 4, instead, provide IV estimates of the impact of total tourism expenditure on total municipal income when controlling for a linear and quadratic time-trends at the local area level, respectively. Standard errors clustered at Local Area level in parentheses. The Table reports instrumental variable estimates. *** p < 0.01, ** p < 0.05, * p < 0.1

| | | | - | | | |
|-----------------------------------|----------------|----------------|----------------|----------------|----------------|----------------|
| | (1) | (2) | (3) | (4) | (5) | (6) |
| | Only | Drop | Pre- | Weight | Weight | Weight |
| | EU 2000 | Provincial | 2013 | Country | Country | Country |
| Estimation: IV | Countries | Capitals | | Distance | Population | GDP |
| | | | | | | |
| Total TE | $21,609^{***}$ | $30,539^{***}$ | $44,793^{***}$ | $25,188^{***}$ | $25,477^{***}$ | $24,669^{***}$ |
| | (1,862) | (4, 471) | (5,049) | (2,790) | (2,813) | (2,609) |
| Observations | 7.020 | 6.858 | 5.070 | 7.020 | 7.020 | 7.020 |
| Municipality FE | ÝES | ÝES | ÝES | ÝES | ÝES | ÝES |
| Year FÉ | YES | YES | YES | YES | YES | YES |
| F-stat of the Excluded Instrument | 71.25 | 50.76 | 57.16 | 48.35 | 48.65 | 52.31 |

| 100100, $1001000000000000000000000000000000000$ | Table 8: | Tourism | and | Economic | Develo | pment - | Robustness | (2) |) |
|---|----------|---------|-----|----------|--------|---------|------------|-----|---|
|---|----------|---------|-----|----------|--------|---------|------------|-----|---|

Note: Dependent variable: total municipal income. Total TE (Total Tourist Expenditure) is reported in thousands of euros. Time horizon 2000-2017 for Column 1-2 and 4-6. Time horizon 2000-2012 for Column 3. Standard errors clustered at the Local Area levels in parentheses. The Table reports instrumental variable estimates. *** p < 0.01, ** p < 0.05, * p < 0.1

| | (1) | (2) | (3) | (4) | (5) | (6) |
|--------------------------------|----------------|--------------|----------------|----------------|----------------|-----------------|
| Panel A: | Total | Taxpayers | Taxpayers | Taxpayers | Taxpayers | Taxpayers |
| Extensive Margin | Taxpayers | below 15K | 16K-25K | 26K-55K | 56K-75K | above 75K |
| | | | | | | |
| Total TE | 0.2359^{***} | -1.4142*** | 0.3235^{***} | 0.4324^{***} | 0.0253^{***} | 0.0316^{***} |
| | (0.0275) | (0.1911) | (0.0489) | (0.0465) | (0.0025) | (0.0031) |
| | | | | | | |
| | (7) | (8) | (9) | (10) | (11) | (12) |
| | Total | Income | Income | Income | Income | Income |
| Panel B: | Income | per taxpayer | per taxpayer | per taxpayer | per taxpayer | per taxpayer |
| Intensive Margin | per taxpayer | below $15K$ | 16K-25K | 26K-55K | 56K-75K | above $75K$ |
| | | | | | | |
| Total TE | 0.0323^{**} | -0.0119 | -0.0159 | 0.0462^{**} | -0.2529^{*} | -1.2297^{***} |
| | (0.0143) | (0.0219) | (0.0106) | (0.0194) | (0.1288) | (0.4528) |
| | | | | | | |
| Observations | 7,020 | 7,020 | 7,020 | 7,020 | 7,020 | 7,020 |
| Municipality FE | YES | YES | YES | YES | YES | YES |
| Year FE | YES | YES | YES | YES | YES | YES |
| F-stat of the Excl. Instrument | 49.60 | 49.60 | 49.60 | 49.60 | 49.60 | 49.60 |

Table 9: Tourism and Economic Development: Extensive and Intensive Margins

Note: Time horizon 2000-2017. Total TE (Total Tourist Expenditure) is reported in thousands of euros. Panel A: Outcome variables refer to the number of taxpayers related to each income class. Panel B: Outcome variables are computed as the ratio between income and number of taxpayers of each income class. Standard errors clustered at Local Area level in parentheses. The Table reports instrumental variable estimates. *** p<0.01, ** p<0.05, * p<0.1

Appendix. Additional Figures & Tables





Note: The graph shows the estimated leads and lags coefficients—with 95% confidence intervals—of the difference in the mean number of tourists (left panel) and in the mean tourist expenditure (right panel) in *Montalbano* municipalities (solid line) and in control municipalities (dashed line) with respect to the timing of the *Montalbano* series release. For the ease of data visualization, the estimated coefficients for control municipalities are plotted slightly on the right with respect to the ones of *Montalbano* municipalities.



Figure A.2: Spillovers in other southern regions

Note: The graph shows the estimated leads and lags coefficients—with 95% confidence intervals—of tourist expenditure in other southern Italian regions with respect to the timing of the *Montalbano* series release. For the ease of comparison with our baseline estimates, the scale of the vertical axis is set between -500 and 2500 as in Figure 6.

| Table A.1: Summary Statistics | | | | | | | | |
|-----------------------------------|-----------|---------------|---|-----------|---------|--|--|--|
| Variable Mean Std. Dev. Min. Max. | | | | | | | | |
| Panel A: Whole Sample | | | | | | | | |
| N. of Tourists | 7.034 | 187.942 | 0 | 61553.078 | 2653560 | | | |
| N. of Nights | 0.126 | 2.736 | 0 | 493 | 2653560 | | | |
| Tourist Expenditure (thousands) | 5.601 | 155.27 | 0 | 42664.191 | 2653560 | | | |
| N. of Tourists (Tripadvisor) | 0.05 | 1.085 | 0 | 208 | 2021760 | | | |
| Panel B: Treated Countries & Tr | reated M | unicipalities | | | | | | |
| N. of Tourists | 99.498 | 700.971 | 0 | 25742.539 | 8064 | | | |
| N. of Nights | 1.371 | 8.009 | 0 | 240 | 8064 | | | |
| Tourist Expenditure (thousands) | 70.697 | 590.988 | 0 | 25163.711 | 8064 | | | |
| N. of Tourists (Tripadvisor) | 1.045 | 5.941 | 0 | 166 | 6144 | | | |
| Panel C: Treated Countries & Co | ontrol M | unicipalities | | | | | | |
| N. of Tourists | 17.775 | 315.451 | 0 | 61553.078 | 778176 | | | |
| N. of Nights | 0.315 | 4.466 | 0 | 493 | 778176 | | | |
| Tourist Expenditure (thousands) | 14.202 | 258.842 | 0 | 42664.191 | 778176 | | | |
| N. of Tourists (Tripadvisor) | 0.118 | 1.861 | 0 | 208 | 592896 | | | |
| Panel D: Control Countries & Tr | reated M | unicipalities | | | | | | |
| N. of Tourists | 7.888 | 138.297 | 0 | 13904.1 | 19152 | | | |
| N. of Nights | 0.149 | 2.14 | 0 | 127 | 19152 | | | |
| Tourist Expenditure (thousands) | 5.71 | 92.023 | 0 | 6460.986 | 19152 | | | |
| N. of Tourists (Tripadvisor) | 0.132 | 0.795 | 0 | 25 | 14592 | | | |
| Panel E: Control Countries & Co | ontrol Mu | inicipalities | | | | | | |
| N. of Tourists | 2.099 | 79.778 | 0 | 28384.32 | 1848168 | | | |
| N. of Nights | 0.04 | 1.412 | 0 | 307.5 | 1848168 | | | |
| Tourist Expenditure (thousands) | 1.694 | 68.762 | 0 | 24654.119 | 1848168 | | | |
| N. of Tourists (Tripadvisor) | 0.016 | 0.256 | 0 | 27 | 1408128 | | | |

Notes: Bank of Italy data refer to the time horizon 1997-2017. *Tripadvisor* data refer to the time horizon 2002-2017. Treated countries are those exposed to the Montalbano series release. Specifically, Finland, France, Germany, Greece, Hungary, Spain, Sweden, and United Kingdom, as reported in Figure 2. Treated municipalities are those in which the series was staged as it is reported in figure 1.

| Variable | Mean | Std. Dev. | Min. | Max. | Ν |
|--|--------|-----------|-------|---------|------|
| | | | | | |
| Panel A: Whole Sample | | | | | |
| Total Income (thousands) | 108293 | 414886 | 1448 | 7370759 | 7020 |
| Total Taxpayers | 7352 | 22234 | 128 | 373757 | 7020 |
| Total Income per taxpayer | 12317 | 2724 | 5473 | 26225 | 7020 |
| N. of Firms in Tourism Sector | 49 | 138 | 0 | 2459 | 5850 |
| N. of Workers in Tourism Sector | 161 | 540 | 0 | 9943 | 5850 |
| Total TE (thousands) | 1981 | 13517 | 0 | 371436 | 7020 |
| | | | | | |
| Panel B: Treated Municipalities | | | | | |
| Total Income (thousands) | 374366 | 275235 | 42974 | 858142 | 72 |
| Total Taxpayers | 26428 | 16309 | 5978 | 51381 | 72 |
| Total Income per taxpayer | 12914 | 2575 | 7189 | 17686 | 72 |
| N. of Firms in Tourism Sector | 204 | 138 | 33 | 521 | 60 |
| N. of Workers in Tourism Sector | 710 | 569 | 70 | 2129 | 60 |
| Total TE (thousands) | 8707 | 14568 | 0 | 72612 | 72 |
| | | | | | |
| Panel C: Control Municipalities | | | | | |
| Total Income (thousands) | 105535 | 415210 | 1448 | 7370759 | 6948 |
| Total Taxpayers | 7154 | 22202 | 128 | 373757 | 6948 |
| Total Income per taxpayer | 12311 | 2725 | 5473 | 26225 | 6948 |
| N. of Firms in Tourism Sector | 48 | 137 | 0 | 2459 | 5790 |
| N. of Workers in Tourism Sector | 155 | 537 | 0 | 9943 | 5790 |
| Total TE (thousands) | 1912 | 13489 | 0 | 371436 | 6948 |

Notes: Bank of Italy data refer to the time horizon 2000-2017. Variables on the number of firms and workers in the tourism sector are built on data from the Bank of Italy, time horizon 2001, and 2004-2017. Data on the number of firms and workers in the tourism sector at the municipality-year level are available from 2004, data for the year 2001 are census data. Balanced panel at the municipality-year level. Total TE (Total Tourist Expenditure) is reported in thousands of euros as well as total personal income. Treated municipalities are those in which the series was staged as it is reported in figure 1.

| | (1) Cohort 1 | (2) Cohort 2 | (3) Cohort 3 | (4) Cohort 4 | (5) Cohort 5 |
|--|--------------------------------|---|--|--|--|
| Panel A: Number of Tourists | | | | | |
| Montalbano mun. \times Country treated | $1,720.212^{***} \\ (172.827)$ | $\substack{1,336.325^{***}\\(129.314)}$ | 352.037^{***} (24.243) | $\substack{1,106.939^{***}\\(18.568)}$ | $\begin{array}{c} 460.752^{***} \\ (10.406) \end{array}$ |
| Panel B: Tourist Expenditure | | | | | |
| Montalbano mun. \times Country treated | $1,207.390^{***}$ (121.627) | $302.874^{***} \\ (27.725)$ | $\begin{array}{c} 144.351^{***} \\ (13.181) \end{array}$ | $\begin{array}{c} 465.538^{***} \\ (12.402) \end{array}$ | 330.525^{***} (3.823) |
| Observations Country-Municipality FE Country-Time FE Municipality-Time FE | 171,990 YES YES YES | 171,990 YES YES VES | 171,990 YES YES YES | 163,800 YES YES VES | 163,800 YES YES YES |

Table A.3: Estimation by Cohorts of treated

Note: Time horizon 1997-2017. Panel data at the country-municipality-year level. Tourist expenditure is reported in thousands of euros. Each cohort of estimation includes only control countries and countries with the same starting-treatment year. The starting-treatment year is defined as the year of the first summer holiday season after the release of the series in country c. That is, it is equal to the release year if the month of release in country c is before September. Instead, it is equal to year after the release if the month of release in country c is after August. Accordingly, in terms of treated countries, Cohort-1 includes France and Sweden (2001), Cohort-2 includes Finland and Germany (2002), Cohort-3 includes United Kingdom and Hungary (2009), Cohort-4 includes Spain (2012), and Cohort-5 includes Greece (2013). Standard errors clustered at Country and Local Area levels in parentheses. *** p<0.01, ** p<0.05, * p<0.1

| | (1) | (2) | (3) | (4) | (5) | (6) |
|-----------------|-------------------------|-----------------------|----------------------|-----------------------|---------------------|---------------------|
| | Total | Income | Income | Income | Income | Income |
| Estimation: OLS | Income | below $15K$ | 16K-25K | 26K-55K | 56K-75K | above $75K$ |
| Total TE | $2,506^{**}$ (1,052) | -539^{***} (161) | -356^{**} (159) | $2,080^{**}$ (804) | 360^{**} (148) | 959^{**} (391) |
| Observations | 7,020 | 7,020 | 7,020 | 7,020 | 7,020 | 7,020 |
| Municipality FE | YES | YES | YES | YES | YES | YES |
| Year FE | YES | YES | YES | YES | YES | YES |

Table A.4: Tourism and Economic Development - OLS

Note: Time horizon 2000-2017. Total TE (Total Tourist Expenditure) is reported in thousands of euros. Standard errors clustered at the Local Area levels in parentheses. Table reports OLS estimates. *** p<0.01, ** p<0.05, * p<0.1